Author's response to reviews

Title: Mobility and cognition are associated with wellbeing and health related quality of life among older adults

Authors:

Jennifer C Davis (jennifer.davis@ubc.ca)
Stirling Bryan (stirling.bryan@ubc.ca)
Linda C Li (lili@arthritiscresearch.ca)
John R Best (john.best@ubc.ca)
Chun Liang Hsu (liang.hsu@hiphealth.ca)
Caitlin Gomez (caitlin.gomez@hiphealth.ca)
Kelly A Vertes (kelly.vertes@hiphealth.ca)
Teresa Liu-Ambrose (tlambrose@ubc.ca)

Version: 5 Date: 10 April 2015

Author's response to reviews:

9 April 2015,

Dear Professor Cameron,

We are pleased to re-submit our manuscript titled, “Mobility and cognition are associated with wellbeing and health related quality of life among older adults” to BMC Geriatrics. The manuscript is not being considered for publication elsewhere.

Thank you for the comments from the Editorial Team. We have been diligent in incorporating the reviewer and editorial team comments. These suggestions are addressed in the ‘Response to Reviewers’ document. We also attached a clean version of the manuscript with ‘Track Changes’ accepted.

We look forward to clarifying further as needed.

All authors have no conflict of interest and financial disclosures to declare.

Yours Sincerely,

Jennifer C. Davis, PhD
Centre for Clinical Epidemiology and Evaluation, 828 West 10th Avenue, University of British Columbia & Vancouver Coastal Health Research Institute (VCHRI), Vancouver, British Columbia, V6T 2B5, Canada, jennifer.davis@ubc.ca

Stirling Bryan
Centre for Clinical Epidemiology and Evaluation, 828 West 10th Avenue, University of British Columbia & Vancouver Coastal Health Research Institute
Linda C. Li  
Arthritis Research Centre of Canada, 5591 No. 3 Road, Richmond BC, British Columbia, V6X 2C7, Canada

John R. Best  
Aging, Mobility, and Cognitive Neuroscience Lab, 2211 Wesbrook Mall, University of British Columbia, Vancouver, British Columbia, V6T 2B5, Canada

Liang Hsu  
Aging, Mobility, and Cognitive Neuroscience Lab, 2211 Wesbrook Mall, University of British Columbia, Vancouver, British Columbia, V6T 2B5, Canada

Caitlin Gomez  
Aging, Mobility, and Cognitive Neuroscience Lab, 2211 Wesbrook Mall, University of British Columbia, Vancouver, British Columbia, V6T 2B5, Canada

Kelly Vertes  
Aging, Mobility, and Cognitive Neuroscience Lab, 2211 Wesbrook Mall, University of British Columbia, Vancouver, British Columbia, V6T 2B5, Canada

Teresa Liu-Ambrose  
Department of Physical Therapy, 2177 Wesbrook Mall, University of British Columbia, Vancouver, British Columbia, V6T 2B5, Canada